

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Pazos, Nicolas; Favara, Marta; Sanchez, Alan; Scott, Douglas; Behrman, Jere R.

Working Paper Long-Term Effects of Rainfall Shocks on Foundational Cognitive Skills: Evidence from Peru

IZA Discussion Papers, No. 15940

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Pazos, Nicolas; Favara, Marta; Sanchez, Alan; Scott, Douglas; Behrman, Jere R. (2023) : Long-Term Effects of Rainfall Shocks on Foundational Cognitive Skills: Evidence from Peru, IZA Discussion Papers, No. 15940, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/272567

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 15940

Long-Term Effects of Rainfall Shocks on Foundational Cognitive Skills: Evidence from Peru

Nicolas Pazos Marta Favara Alan Sánchez Douglas Scott Jere Behrman

FEBRUARY 2023



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 15940

Long-Term Effects of Rainfall Shocks on Foundational Cognitive Skills: Evidence from Peru

Nicolas Pazos University of Nottingham

Marta Favara University of Oxford and IZA

Alan Sánchez Grupo de Análisis para el Desarrollo

FEBRUARY 2023

Douglas Scott University of Oxford

Jere Behrman University of Pennsylvania

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Long-Term Effects of Rainfall Shocks on Foundational Cognitive Skills: Evidence from Peru¹

Global warming is changing precipitation patterns, harming communities strongly tied to agricultural production, particularly in low-and-middle income countries (LMICs). Whilst the long-term effects of being exposed to rainfall shocks early in life on school achievement tests are well-established, there is little population-based evidence from LMICs on the mechanisms through which these shocks operate. This paper analyses the effects of early exposure to rainfall shocks on four foundational cognitive skills (FCSs), including executive functions (EF) that have been found to be key predictors of educational success. These skills were measured via a series of tablet-based tasks administered in Peru as part of the Young Lives longitudinal study (YLS). We combine the YLS data with gridded data on monthly precipitation to generate monthly, community-level rainfall estimates. The key identification strategy relies on temporary climatic shocks being uncorrelated with other latent determinants of FCS development. Our results show significant negative effects of early life exposure to rainfall shocks on EF. We also find evidence of rainfall shocks decreasing households' abilities to invest in human capital, which may affect both FCS and domain-specific test scores. Interestingly, social policies providing affected households with additional resources partially offset the effects of the rainfall shocks.

JEL Classification:J24, Q54, I24, I14Keywords:skills formation, human capital, rainfall, Peru, early childhood

Corresponding author:

Marta Favara Oxford Department of International Development University of Oxford 3 Mansfield Road, OX1 3TB Oxford United Kingdom E-mail: marta.favara@qeh.ox.ac.uk

¹ This study was funded by the Eunice Kennedy Shriver National Institute of Child Health and Human Development (NICHD R21 HD097576) research grant on "Foundational cognitive skills in developing countries: early-life nutritional, climatic and policy determinants and impacts on adolescent education, socio-emotional competencies and risky behaviors". The funder had no role in the design, interpretation or writing-up of the study or in the decision to submit the study for consideration for publication. Thanks also to the UK's Foreign, Commonwealth and Development Office (FCDO) for funding Young Lives at Work and enabling this research (Grant no. GB-GOV-1-301108). We are grateful to Santiago Cueto, Richard Freund, Annina Hittmeyer, Jennifer López, Margaret Sheridan and participants at the LACEA conference (Lima, 2022) for providing very useful comments.

1. Introduction

Despite considerable progress in recent years, undernutrition among children is still highly prevalent in low- and middle-income countries (LMICs). An estimated 249 million children under 5 years of age are at risk of not being able to realize their developmental potential, with the primary indicator for 170 million of these being undernutrition, as evidenced by stunting (Black et al., 2017; Lu et al., 2016). Climate change is partially responsible for slowing down the long secular declines in undernutrition, as it is rapidly increasing weather variation causing more unpredictable precipitation patterns (USGCRP, 2017). This unpredictability especially affects LMIC communities that are economically tied to agricultural production and that often lack access to good irrigation and water-management systems. Children from these communities are particularly vulnerable in their early years to economic shocks, which often have long-term consequences for their human-capital formation (Almond et al., 2018).

The long-term implications of early life exposure to rainfall shocks on health and nutrition (Dimitrova & Muttarak, 2020; Randell et al., 2020; Hirvonen et al., 2020; Skoufias & Vinha, 2012), and on school domain-specific achievement tests and educational attainments (Maccini & Yang, 2009; Thai & Falaris, 2014; Randell & Gray, 2019; Hyland & Russ, 2019) are fairly well-established.² However, to our knowledge, there is little population-based evidence from LMICs about the mechanisms through which climate shocks may lead to poorer domain-specific achievement-test performance and educational outcomes, making it difficult to inform remediating policies.

This paper contributes toward addressing this gap by investigating the long-term impacts of exposure to rainfall shocks during the first 1,000 days of life on foundational cognitive skills (FCS), which may be important mechanisms through which climate shocks could affect domain-specific achievements. We use unique data collected through the administration of series of tablet-based tasks included in the Rapid Assessment of Cognitive and Emotional Regulation, RACER (Behrman et al., 2022; Hamoudi & Sheridan, 2015). We collected these data as part of the Young Lives Study (YLS), one of only a handful of LMIC cohort studies that can be used for life-course analysis over infancy, childhood and adolescence, and the only one with data on FCS, rarely available in large sample surveys in LMICs. RACER includes tasks to measure inhibitory control (i.e., the capacity to control attention or behaviour and override counterproductive impulses); working memory (i.e., the capacity to hold in mind and manipulate information not visible in the environment); long-term memory (i.e., the capacity to encode, retain, and retrieve information) and implicit learning (i.e., the capacity to learn without conscious awareness).

² Skills such as reading, arithmetic, or linguistics, are domain-specific, as they relate to specific types of knowledge. In contrast, domain-general skills, such as inhibitory control or working memory, are linked to the accumulation of knowledge across multiple domains (Hamoudi & Sheridan, 2015).

Both inhibitory control and working memory are components of executive functioning (EF), a specific group of mental capacities required to formulate goals, plan how to achieve these goals and carry-out these plans effectively (Lezak, 1982). EF is increasingly found to be a key domain of child development and a key predictor of educational success. For example, some studies link EF to school readiness, more so than IQ, entry level reading or math skills (Fernald et al., 2009; Blair and Razza, 2007; Blair, 2002). Other studies find that individuals with higher EF are less likely to engage in risky behaviours related to health and crime. Furthermore, the existing evidence shows that EF is highly associated with early-life household socio-economic status and with investments by parents and teachers (Noble et al., 2005, 2007; Farah et al., 2006; Klenberg et al., 2001; Ardila et al., 2005), suggesting that EF is malleable, at least during childhood, and not predetermined at birth. Stressful, challenging, or deprived conditions may impede the development of these skills and hasten their decay (Lupien et al., 2009; Shonko & Garner 2012; Nelson & Sheridan, 2011; McLaughlin et al, 2014; Sheridan & McLaughlin, 2014; Sheridan et al., 2012, 2013). The periods of development span well into the second decade of life for working memory and inhibitory control, both of which are key elements of EF.

Many of the existing studies on EF or, more generally, FCS, have focused on high-income countries (HICs), as opposed to LMICs, where arguably children are at greater risk of not reaching their developmental potential and wherein live the vast majority of the world's children (Black et al., 2017; Lu et al., 2016). Furthermore, most evidence comes from experimental studies with (often very) small samples. Climate shocks, which have been shown to have an impact on schooling achievement, might also have an impact on FCS and EF. However, to our knowledge, this potential connection is yet to be analysed.

For our analysis, we combine the YLS child-level data with gridded estimates of monthly total precipitation across land surfaces from the Matsuura and Wilmott (2015) dataset. We use the grid points for the YLS communities to create monthly, community-level rainfall estimates and construct the Standard Precipitation Index (SPI) (Lloyd-Hughes & Saunders, 2002). We adopt the conventional measure of rainfall shocks as a deviation of rainfall from the historical averages, during the crop-growing rainy season (defined as the four months with the highest average rainfall, at the community level), for the same locality, and during the same period.

We exploit rainfall fluctuations to investigate whether: (i) the exposure to climatic shocks early in life negatively affects FCS measured in adolescence; (ii) the magnitude of the impact on FCS depends on when the shock was experienced, either during the gestational period or the first two years of life, and on the nature of the shock (abnormally high or abnormally low rainfall). Further, we investigate some potential mechanisms through which the shocks could have affected FCS, by (iii) testing whether the negative effects of shocks are stronger for children growing-up in agricultural households as opposed

to households that do not rely on agriculture as the main source of income, (iv) exploring potential effects on other indicators of human-capital accumulation (mainly height-for-age and vocabulary) for which we have data from earlier in their lives; and (v) analysing the effects on households' food security. Finally, (vi) we explore the potential remediation role of social policies by analysing heterogeneous effects on children from the JUNTOS conditional-cash-transfers (CCT) program.

The key identification strategy relies on the assumption that temporary climatic shocks, are uncorrelated with other latent determinants of FCS development. Our identification strategy uses community fixed effects to control for unobserved time-invariant characteristics common to all children in the same community, alongside year- and month-of-birth fixed effects to control for events that might have an impact across all children born during the same year or in specific calendar months. Importantly, our strategy relies on the presence of data for sibling pairs, providing additional variation in the timing and nature of early life rainfall shocks: the YLS index children who were all born over a 12-month period and were between 12 and 13 years old at the time RACER was implemented, and their younger siblings who were between 6 and 12 years old at this time.

Our results show robust negative effects of rainfall shocks on FCS and particularly on working memory and inhibitory control. The gestational period is particularly sensitive to these shocks, with abnormally low and high rainfall shocks appearing to affect FCS equally. Furthermore, our results provide some evidence on the reduction of households' incomes as a potential mechanism through which rainfall shocks affect FCS. Specifically, rainfall shocks seem to affect the abilities of households to invest in their children's nutrition. We hypothesise that these changes in income are related to reductions in agricultural output in the region, although we do not find any differentiated effects for agricultural households. Finally, we find evidence suggesting that there is space for public policies providing monetary relief to remediate for the negative effects of rainfall shocks on FCS.

The rest of the paper is organized as follows: section 2 briefly summarizes the existing literature on the impact of rainfall shocks on educational attainment and domain-specific learning outcomes; section 3 outlines the conceptual framework; section 4 and section 5 describe, respectively, the data and the empirical strategy used for the study; section 6 presents the main results; section 7 includes an exploration of some of the potential mechanisms behind the relationships uncovered between rainfall shocks and FCS; section 8 concludes.

2. Literature review

2.1 The impact of rainfall shocks

The long-term implications of early life exposure to rainfall shocks on educational attainments are fairly well-established. Maccini & Yang (2009) evaluate the impact of rainfall deviations (from local means) during early childhood in Indonesia, and find that higher rainfall during the year of birth leads to higher schooling attainment for women, but no effects for men, nor for shocks after or prior to birth years. The authors argue that the result is driven by the positive impact of rainfall on agricultural output. Similarly, Thai & Falaris (2014) find that higher rainfall (above the commune-of-birth average), during the year of birth and the third year of life, shortens school-entry delays and enables faster progression through school in Vietnam. Randell & Gray (2016) find that an increase in rainfall during the summer agricultural season in rural Ethiopia is associated with an increase in the probability of completing at least one year of school and attending school at the time of the interview for children affected during their first seven years of life. Hyland & Russ (2019) use DHS data for women in 19 countries in sub-Saharan Africa and find that being exposed to droughts (measured as a Standardized Precipitation Index, SPI lower than -1.5) during early childhood has negative effects on schooling attainment, adult heights and wealth (for women from rural households only). Some studies also note that there is a nutrition-learning link that is triggered by climate shocks. For example, Alderman et al. (2006) use droughts as an instrumental variable to measure the effects of height-for-age on school grades completed.

Rainfall shocks do not affect everyone equally. Some studies only find effects for women (Maccini & Yang, 2009) and others find effects on rural households only (Hyland & Russ, 2019). These mixed results reflect different contexts and channels through which climate shocks affect human-capital accumulation. It is argued that exposure to rainfall shocks during the school period could lead to positive effects on education if it helps shift resources to increase investment in children's human capital. Shah & Steinberg (2017) find that, for children aged 5 to 16, droughts in India affect children's educational attainment positively, as—they argue—they push children away from agricultural work. Randell & Gray (2019) present another case of mixed results. Using census data for 29 countries in the Global South, they find that experiencing droughts during the gestational period and in early childhood is associated with lower schooling attainment for children in Southeast Asia, and northern and central Africa, but associated with more schooling attainment for children in Central America and the Caribbean.

There is less abundant evidence on the impact of climate shocks on cognitive achievement, as generally measured by domain-specific test scores. Rosales-Rueda (2018) exploits the *El Niño* phenomenon in Ecuador to show evidence of the effects of early life exposure to excessive rainfall (floods) on the

Peabody Picture Vocabulary Test (PPVT). More recently, Chang et al. (2022) use the YLS data to evaluate the long-term effects of in-utero rainfall shocks (droughts and floods) on domain-specific achievement tests in India. They find negative long-lasting effects on math (at age 15), but only short-term effects on vocabulary as measured by PPVT.

Overall, the existing literature uncovers a negative effect of rainfall shocks on schooling attainment and domain-specific cognitive-achievement test scores, although the magnitudes (and sometimes the directions) of the effects vary significantly and differ across different population subgroups. However, evidence on the impact of climate shock on domain-general cognitive skills is lacking. Furthermore, with few exceptions (Rosales-Rueda, 2018; Chang et al., 2022), the literature has emphasized the role of droughts, either explicitly (e.g., Hyland & Russ, 2019) or implicitly by analysing the role of the total amount of rainfall — but not excessive rain — in countries and/or time periods where the population is prone to droughts (e.g., Maccini & Yang, 2009).

2.2 How climate shocks affect cognitive skills

The existing literature suggest that climate shocks may affect cognitive skills through two main mechanisms: directly by affecting cognitive functioning (for example altering brain chemistry, electrical properties and function) or indirectly by affecting human-capital-accumulation processes (altering investments in relevant inputs of human-capital production). Most evidence on the direct effects comes from the medical literature,³ while evidence on indirect effects comes predominantly from the economic literature, and documents the effect of environmental shocks, in terms of child health, nutrition and cognition (Dercon & Porter, 2014; Berhane et al., 2014; Baez et al., 2009). This paper contributes to this second stream of literature and the rest of this section briefly summarizes the main related evidence.

One clear way in which rainfall shocks could affect investment in human-capital accumulation is through income. An excess or lack of rainfall could have impacts on agricultural output (Felkner et al., 2009), which could directly affect the income and consumption of agricultural households (Dercon, 2004; Jayachandran 2006; Amare et al. 2018). However, non-agricultural households could also be impacted (Rijkers & Söderbom, 2013; Grabrucker & Grimm, 2021), as a decrease in agricultural outputs could increase food prices, or decrease food availabilities in markets and the intakes of important nutrients. Additionally, abnormally high or low levels of rainfall could affect other sources of income either directly linked to agriculture and food prices (wholesale and retail trade, transportation,

³ For example, studies suggest that elevated temperatures affect completing complex cognitive tasks (Fine & Kobrick, 1978; Hocking et al., 2001; Park, 2017) and diminish attention, memory, information retention and processing (Vasmatzidis et al., 2002)

restaurants), or not directly linked to agriculture (i.e., construction, destruction of infrastructure), and through changes in the general equilibrium (i.e., less money spent on services).

Overall, decreases in households' incomes may directly affect children's nutritional intakes, which, in turn, may limit healthy development and growth (Nicholas et al., 2021; Dimitrova & Muttarak, 2020; Cornwell & Inder, 2015; Skoufias & Vinha, 2012). The existing evidence suggests that poor nutritional status is associated with lower FCS (Sánchez et al., 2022). These impacts are particularly relevant during the first 1,000 days of life, a period in which shocks are more prone to affect the long-term acquisition of cognitive skills, as widely recognized in the nutritional and economic literature (Victora et al., 2008; Maluccio et al., 2009; Victora et al. 2010; Almond et al., 2018).

3. Conceptual framework

Following closely Todd & Wolpin (2003), and Glewwe & Miguel (2007), we consider a framework in which the first 1,000 days (the in-utero period plus the first two years after birth) is taken as period t=1, and the remainder of childhood as period t=2. In period t=1, family of child *i* can invest in cognitive skills ($S_{1,i}$), as follows:

$$S_{1,i} = f_1(F_{1,i}^S, F_{1,i}^H, S_{0,i}) \tag{1}$$

Where $F_{1,i}^S$ denote family inputs of an educational nature in period 1; $F_{1,i}^H$ are health-related family inputs in period 1; and $S_{0,i}$ is the cognitive endowment. Following an analogous structure, cognitive skill in period 2 ($S_{2,i}$) can be expressed by the history of inputs invested in the child:

$$S_{2,i} = f_2(F_{2,i}^S, F_{1,i}^S, F_{2,i}^H, F_{1,i}^H, S_{0,i})$$
(2)

Here, $F_{2,i}^S$ and $F_{2,i}^H$ represent investments in period 2. For simplicity of exposition, we assume parents are altruistic so that they gain utility from their children's cognitive skills in addition to their own consumption.⁴ The utility function for family *i* can be expressed for the one-child case as:

$$U_i = f(C_{1,i}, C_{2,i}, S_{1,i}, S_{2,i})$$
(3)

where $C_{t,i}$ is parental consumption in period *t*. Resources to pay for family inputs and consumption in each period *t* come from household income in that period $(I_{t,i})$ with, for simplicity, no transfer of

⁴ Alternatively, or in addition, parents may be interested in their children's cognitive skills because the expected support the children can provide to the parents in their old age increases in their children's skills.

budgetary resources across periods. We assume household income is negatively correlated with rainfall shocks (θ), and the prices of inputs p are positively correlated with rainfall shocks, under the usual assumption that markets in LMICs tend to be segmented and only weakly linked to national and international markets.

$$I(\theta)_{t,i} = p(\theta)_{c,t} C_{t,i} + p(\theta)_{H,t} F_{t,i}^{S} + p(\theta)_{S,t} F_{t,i}^{H}$$
(4)

Within this framework, parents are expected to choose optimally for inputs $F_{t,i}^{S}$, $F_{t,i}^{H}$, and consumption $C_{t,i}$ over t = 1, 2, subject to constraints (1) to (4). This framework is highly simplified (for example, the roles of parental time and community inputs are not incorporated) but seeks to illustrate that, as is commonly observed in poor LMIC households, household incomes and input prices are prone to be affected by rainfall shocks. For this reason, the occurrence of rainfall shocks can lead to reduced investment in cognitive skills in period 1, which are transferred to the next period by the inclusion of lagged inputs in the cognitive production function for period 2 in equation 2.

Of course, this simple model does not consider that households may have ways of coping with shocks. For example, households could use their savings, sell assets (i.e., livestock) or access credit markets to mitigate short-term shocks. In the context of LMICs, poor households usually have little savings and face constraints to access credit markets. However, it is possible that strategies of this type are used to mitigate the effects of the shocks and, sometimes, can successfully limit some of the negative effects of rainfall shocks on cognitive skills. Thus, it is possible that the effects observed in our study are less pronounced than they otherwise would be due to these strategies.

Other mitigation strategies, particularly relevant in the context of households involved in agriculture, include finding other sources of income outside of the agricultural sector, or investing more time working at the farm. These other strategies could still result in decreases in inputs for cognitive skills, as they could force caregivers into the labour market at times that are crucial for children's development. In terms of the model, caregivers' hours spent working implies fewer hours invested in providing inputs into child development, $F_{1,i}^S$ or $F_{1,i}^H$.

All of these strategies have the peculiarity that, even if they help mitigate the negative effects of the shock by increasing income $I(\theta)_{t,i}$, they do nothing to reduce the size of the impact θ , which might still affect prices. However, there are some other coping mechanisms that could lessen the effects of rainfall shocks on input and consumption prices. In the case of higher agricultural input prices created by shocks that reduce agricultural outputs, agricultural households could potentially mitigate these effects by shifting their production to more resilient products (either more water-resistant or less water-

dependent). This is only possible in a given crop year if the rainfall shock arrives before the beginning of production. To account for this possibility, we are only considering rainfall shocks happening during the rainy season, which works as a proxy for the period in which most crop-choice decisions should have already been made⁵. Similarly, we consider the intensity of the shocks by measuring the total number of months in which shocks were experienced during rainy seasons, as multiple rainfall shocks should be harder to prevent or mitigate.

4. Data

4.1 The Young Lives study

The YLS is a longitudinal research study on childhood poverty that follows 12,000 children in four countries: Ethiopia, India (Andhra Pradesh and Telangana), Peru and Vietnam. The first survey took place in 2002 with four further rounds of in-person data collection in 2006/07 (Round 2), 2009/10 (Round 3), 2013/14 (Round 4) and in 2015/2016 (Round 5). For the empirical analysis, we use a sample of 1,511 children (index children) who were born in Peru in 2001–2002 and were aged between 6 to 18 months in the first survey round. In addition, we use information on a sample of 605 younger siblings, aged 2 to 8 years in 2009/10, the first time they were interviewed. The RACER tool designed to measure FCS was administered to both cohorts in Round 4 (2013/14) only.

In all rounds, a child questionnaire, including data on child health, anthropometrics and education, and a household questionnaire, including data on caregiver background, livelihood, household composition, socio-economic status, shock were administered. In addition, test scores in vocabulary and math were collected since Round 2 (2006/07). In the siblings' sample, key characteristics, such as anthropometrics and test scores, were collected since Round 3 (2009/10).

The index children were enrolled from 74 communities within 20 selected districts that were randomly selected from the whole universe of Peruvian districts (excluding the wealthiest 5%). After districts were chosen, a community or housing block was randomly selected, and then all dwellings from each cluster of houses were visited to look for children of the right ages. Once a block was completely examined, and if the desired sample size was not met, the next available neighbouring block was visited by the fieldworker, following the same process. This process was followed until the desired sample size was met. The final sample represents the poorest 95% of children from both urban and rural areas in

⁵ It is worth noting that there are other production decisions that could be made after the crop-choice and could potentially mitigate the effects of the shock, mainly regarding the use of certain inputs such as fertilizers. For agricultural households, these decisions imply a reallocation of the household's resources to buy such inputs, which could in turn still affect the amount resources assigned to the child's development.

the three main geographical regions: coast, highland (altiplano) and jungle. More information on the sampling strategy can be found in Escobal et al. (2003).

Although the YLS samples are not nationally representative, comparisons with DHS data show they cover the diversity of children in Peru (Escobal & Flores, 2008). The proportion of rural households is 55% and average mothers' schooling attainment is less than seven grades. The attrition rate across the first four survey rounds that we use is 6.3%, which is relatively low compared to other longitudinal studies. To achieve such low levels of attrition, the study successfully followed most sample families that moved within the country (Sánchez & Escobal, 2020). Table 1 presents descriptive statistics of the sample.

INSERT TABLE 1 HERE

4.2 Measurements of FCS

Data on FCS were collected for both index children and their siblings during Round 4 (2013). At that time, the index children were aged 11 to 12 years, whereas the younger siblings were aged 5 to 11. The FCS were measured using a series of computer-based tasks through Rapid Assessment of Cognitive and Emotional Regulation (RACER) (Hamoudi & Sheridan, 2015; Ford et al., 2019; Behrman et al. 2022).⁶ RACER is a novel touch-screen computer/tablet application that uses five short tasks (one to four minutes each) to assess four components of FCS (Inhibitory Control- IC, Working Memory- WM, Long-term Memory- LTM, and Implicit Learning- IL) in children aged 6 years and older (and adults). RACER was administered to 98.5% of the YLS sample available for interviews. Each assessment included in RACER is designed to be as domain-general as possible (e.g., performance should not be affected by literacy or numeracy) and at the same time as skill-specific as possible (Hamoudi & Sheridan, 2015). To achieve this goal, in each case performance is measured though a baseline task and a challenge task. The former is affected by factors such as the ability to understand instructions or familiarity with the use of digital devices, whereas the latter is designed to measure the same factors and the specific skills of interest (thus, to measure a specific skill, the baseline task results need to be controlled).

Table 2 summarizes the baseline and challenge tasks calculated in each case, and the average performance. LTM is measured by the proportion of correct answers, IC and WM by response times, and IC by a linear combination of response time and accuracy. Children who perform better in LTM

⁶ More information about RACER and the FCS measured in the YLS can be found in Hamoudi and Sheridan (2015) and Behrman et al. (2022).

are expected to have a larger proportion of correct answers, while in the IC, IL or WM tasks children who perform better are expected to require less time. Similarly, those that perform better in WM would provide more accurate answers. To facilitate the interpretation of the results (such that a higher score is linked to better performance) we use an inverse function for our measurements of IC, IL and WM, and all variables are standardized by age in years.⁷ This transformation allows us to directly interpret higher values as higher skills. In the case of the baseline tasks, they differ depending on the outcome variable. For IC, the baseline task consisted of a linear combination of two variables with the same weight (0.5): the average Euclidean distance for all same-side trials and the average response time for same-side trials. In the case of WM, LTM and IC, the baseline trials are composed of only one component, as described earlier. More information about the administration of RACER in the YLS sample can be found in Behrman et al. (2022).

INSERT TABLE 2 HERE

4.3 Rainfall data

Rainfall data are obtained from the University of Delaware, a commonly used climate dataset (see, for instance, Shah & Steinberg (2017), Rocha & Soares (2015), and Thai & Falaris (2014)). These data contain gridded estimates of monthly total precipitation across land surfaces between 1900 and 2017 (Matsuura & Wilmott, 2015). Each of the values is a local point estimate at a 0.5-degree of longitude-latitude resolution, based on publicly available observation-station records.

The rainfall data are used to compute the monthly historical means in each of the 82 YLS communities in Peru.⁸ Using the grid points for which rainfall data are available, the rainfall data are matched to the GPS locations of the YLS communities (using the main square in each community as the reference point) (Mcquade & Favara, 2022).⁹

For each community, the monthly rainfall precipitation is calculated as a distance-weighted average of the monthly rainfall registered at the four closest grid points to that community. The weights used are computed as the inverse of the distances of the grid points from the community over the total inverse distance of all four grid points. This approach puts greater significance on grid points closer to the community.

⁷ For standardization, we use the combined sample of all children in Ethiopia and Peru.

⁸ Communities within YLS are defined as administrative areas, as these are geographically well-defined and recognised areas.

⁹ For any communities that do not have main squares, another reference points are used, such as schools, churches, or post offices.

We then use these data to create the Standardized Precipitation Index (SPI), following Lloyd-Hughes & Saunders (2002). The main reason for using SPI data instead of direct rainfall data is because rainfall data are not usually normally distributed, as they are naturally truncated when reaching zero litters of rain. The SPI normalises the data, making both positive and negative shocks equally represented. We calculate the SPI for every community using data from the last 50 years. We use a 2.0 standard deviation from the historical means to define a rainfall shock.

As discussed, a potential channel through which climate shocks could affect FCS is through impacts on agricultural outputs, which, in turn, may have impacts on both agricultural and non-agricultural households. For this reason, our analysis focuses on shocks during the months in which rainfall variation might have had the largest effect on agricultural outputs. Typically, the rainy-season months are key for agriculture, as a big part of the agricultural outputs destined for national consumption is produced by small farmers with limited access to irrigation and heavily dependent on rain. Therefore, we only consider those rainfall shocks that occur during rainy-season months, defined as the four months with the highest average monthly rainfall in each community.

4.4 Defining rainfall shocks in the pre-natal and early childhood period

We initially define a shock as any monthly SPI deviation of at least 2 standard deviations above (*abnormally high rainfall shocks* or *floods*) or below (*abnormally low rainfall shocks* or *droughts*) the historical monthly average for the same community during each rainy-season month. The main specification of this study does not distinguish between abnormally high or abnormally low shocks, under the assumption that both an excess or a shortage of rainfall, when unpredicted, should negatively impact agricultural production. Here, we assume previous decisions and investments have been made with the expectation of average rainfall for that community. Later, we relax this assumption and we test the heterogenous impact of floods and droughts.

Information about the date and place of birth of each YLS child is used to identify the first 1,000 days of life, critical for human-capital development, and to distinguish between rainfall shocks that occurred during the gestational period (in-utero and/or pre-natal) and shocks that occurred during the early childhood period (the first and second years after birth).

The date of conception and the gestational period of each YLS child is defined using their date of birth and assuming 38 weeks (266 days) as an approximation of a normal-term pregnancy, as per the World

Health Organization definition.¹⁰ Under these assumptions, the index children in our sample were conceived between March 2000 and October 2022, while the younger siblings were conceived between September 2002 and March 2007. We do not account for premature births when defining the gestational period. However, our estimations include (as a control) an indicator variable capturing whether a child was born prematurely, based on self-reported information from the children's mothers. In total, 25% of the children were born prematurely. Shocks happening during the in-utero period occurred from one to nine months before the month of birth. Our definition of gestational period encompasses the whole in-utero period for most children; but might include some additional time for those born prematurely. Hence, we are not making a difference between the in-utero period and the gestational period, and simply defining them as the same time-period. For the early childhood period, we consider shocks during the month of birth and the following 23 months as shocks during the first and second years of life. As we are only considering shocks occurring during the rainy season months, the maximum number of shocks during these periods is four for the in-utero period, and eight for the early childhood period.

4.5 Sample definition

To accurately impute exposure to rainfall shocks during the first 1,000 days given the available data, we restrict the sample to households that had not moved from the community where the index child was born before Round 3 (2009), which accounts for 79% of the sample. To do this, we proceed as follows: First, in order to identify the community of residence of the mother during the gestational period, we use Round 1 information on where the child was living at the time of the interview and information about how long the mother has been living in the same community.¹¹ Second, to exclude mothers who may have migrated to the Round-1 community of residence to give birth (or after the birth of the child), we exclude from the sample any children whose mothers reported living in the Round-1 community for less than 9 months prior to the interview. Third, we restrict the sample to children from families that did not migrate up to Round 3. This allows us to define the community of residence during the relevant periods. We discuss potential issues of sample selectivity in Section 8.

The final subsample of non-migrant families from agricultural communities includes a total of 2,656 children, with 1,511 index children and 610 younger siblings. The final sample covers 82 communities, distributed across all three natural regions (coast, highland, and jungle).

¹⁰ The World Health Organization defines preterm as giving birth before 37 weeks of pregnancy are completed. See the WHO website: <u>https://www.who.int/news-room/fact-sheets/detail/preterm-birth</u>. However, most papers use 266 days or 38/40 weeks as the threshold to define pre-term births.

¹¹ For younger siblings, we use information from Rounds 2 (2006) and 3 (2009) to guarantee that children did not move during their in-utero period and early childhood.

4.6. Rainfall shocks in the Peruvian Young Lives sites

Table 3 presents descriptive statistics for the number of shocks experienced by the children in our sample. We distinguish by type of shock (all shocks, abnormally high rainfall shocks and abnormally low rainfall shocks), period of exposure (first 1,000 days, in-utero period and first two years of life), for all children, index children and younger siblings.

INSERT TABLE 3 HERE

As observed, abnormally high shocks are more prevalent than abnormally low shocks during the relevant time periods. More shocks are experienced during the post-birth period than during the gestational period. This is an expected result, as the post-birth period rainy season includes eight months over 24 months, whereas the in-utero period rainy seasons include only between one and four rainy-season months.

It is also worth noting that there are many children that never experience a shock. To have a better idea of how prevalent exposure is and how the shocks are distributed, Table 4 reports the percentages of children who experience at least one shock and the average number of shocks for those experiencing at least one shock.

INSERT TABLE 4 HERE

In total, 27% of children experience at least one shock during the first 1,000 days of life. For this 27%, we observed that on average children experience almost two shocks during their first 1,000 days of life. This contrasts with the average presented in Table 4. Most children subjected to at least one shock experienced an abnormally high shock, especially during the first or second year of life. As with the averages, the proportion of index children experiencing shocks is higher than the proportion of siblings. In particular, only a minority of the siblings experienced at least one shock within the in-utero period.

5. Empirical Strategy

The main specification (1) used to estimate the effects of rainfall shocks on FCS is:

$$Y_{i,j,y,m} = \alpha_0 + \alpha_1 B_{i,j,y,m} + \alpha_2 Shock_1000 days_{j,y,m} + \beta X_{i,j,y,m} + \varphi R_{i,j,y,m} + \delta_j + \gamma_y + \theta_m + \mu_{i,j,y,m}$$
(1)

Where the dependent variable $Y_{i,j,y,m}$ is a generic variable denoting performance in challenge tasks IC, WM, LTM or IL for child *i* in cluster *j*, born in year *y* and month *m*; $B_{i,j,y,m}$ is the performance of the same child in the baseline task; $Shock_{1000} days_{j,y,m}$ is a variable denoting the number of rainyseason months during the first 1,000 days of life in which a rainfall shock was experienced; $X_{i,j,y,m}$ is a vector of time-invariant controls at the child/household level, $R_{i,j,y,m}$ is a vector of variables relating to the timing of the RACER tests, also defined at the child level, δ_i are cluster-level fixed effects, γ_v and θ_m are year and month-of-birth fixed effects, respectively; and $\mu_{i,j,y,m}$ is the error term. The α_2 term is the coefficient of interest, which denotes the impact of being exposed to extreme rainfall shocks on FCS development. Additionally, we estimate the parameter α_1 , and the vectors containing parameters β and φ . The vector $X_{i,j,y,m}$ contains an indicator variable for the child's sex, a variable for grade of schooling completed by the parent with the higher schooling attainment, an indicator of whether the child was born prematurely, and if the house is an agricultural household. We also include an indicator variable controlling for ethnicity. The vector $R_{i,j,y,m}$ includes indicator variables taking the value 1 if the RACER test was taken on a weekend, if the test was taken between 5pm and 12am, or if the test was taken between 1 and 8am. The δ_i cluster fixed effects control for unobserved (timeinvariant) characteristics common to all children in cluster j; and the γ_y and θ_m year and month-of-birth fixed effects control for all common, time-specific unobserved characteristics that affect children in our sample born in the same year or month.

We will use two variations of this specification in our analysis. First, we estimate results as described, where the shock variable considers all shocks (droughts and floods) experienced during the first 1,000 days. Second, we explore whether there are key characteristics of the shocks that make them more (or less) likely to affect the development of FCS. To do this, we distinguish between abnormally high (floods) and abnormally low (droughts) rainfall shocks that happened during the pre-natal period and the post-natal period, separately.

Given that shocks are defined as monthly standard deviations from the historical means within each community, experiencing a shock during the first 1,000 days could be considered a random event, when holding the historical mean rainfall at the community-level constant. Our main identification strategy requires using three different fixed effects. First, we use community fixed effects to control for unobserved time-invariant characteristics common to all children in the community. Second, we incorporate year-of-birth fixed effects to control for yearly events that could similarly affect all children born during the same year, and finally we add a month-of-birth fixed effects to control for common effects experienced by all children born in the same calendar month.

Furthermore, our strategy significantly relies on the presence of the data for the younger siblings, as they provide much larger variation in terms of years of birth and timing of the shocks. Also, by including year-of-birth fixed effects and ages in months when the RACER tasks were taken, we are indirectly controlling for whether the child is an index child or a younger sibling. It is also important to control for age in months at which the RACER test was administered, as this may directly influence the RACER results.

Variables that control for child gender, parental schooling, mother's ethnicity, and if the child was born prematurely are included as basic controls, as they could affect results in the RACER tasks. In our data, ethnicity is approximate using the mother's language, and it consists of an indicator that takes the value of 1 if the mother's native language is Spanish, and 0 otherwise (implying it is an indigenous language). We also control for cases where the household is an agricultural household, as we expect this may be relevant in how the shock affects the results. For all of our specifications, standard errors are clustered at the district level¹².

As part of our study, we also conduct analysis to explore the different mechanisms through which rainfall could impact FCS, such as being an agricultural household, nutrition, early effects on other cognitive skills, and food security. We also include an analysis of the potential role of public policies for remediating the rainfall shock effects. A more detailed description of these additional analyses, the methodology used, and their results can be found in Section 7.

6. Main Results

Table 5 shows the estimates for our main specification considering the number of shocks happening during the first 1,000 days. All results have standard controls; community, year-of-birth and month-ofbirth fixed effects, controls relating to the administration of the RACER application, and results for the baseline tasks, as defined in the previous section. All estimations include standard errors clustered at the district level.

Columns 1, 2, 3 and 4 report the effects of the shocks on IC, WM, IM and LTM, respectively. We find that each rainfall shock occurring during the first 1,000 days of life significantly decreases WM by 6.6% of a standard deviation. The other point estimates are statistically insignificant.

INSERT TABLE 5 HERE

¹² Communities (as defined in the sample) are in some cases located very close to each other (within the same district), which could lead to spatial correlation. In order to account for this possibility, we cluster the standard errors at the district level.

In Table 6, we report the estimates when distinguishing the effects of abnormally high or abnormally low shocks that happened during the pre-natal and post-natal periods. Our results confirm a negative effect of experiencing an additional rainfall shock on WM (Column 2) and IC (Column 1). While the estimated coefficients are in general larger for droughts than for floods, and during the gestational period, we are unable to reject the null hypothesis that these effects are similar. This might be due to coefficients being imprecisely estimated. No significant effects are found on IL or LTM, at any of the levels considered.

INSERT TABLE 6 HERE

7. Exploring mechanisms

7.1. Heterogeneous effects for agricultural households

One of the main drivers through which rainfall shocks occurring during the first 1,000 days could affect the development of cognitive skills is through the impact of shocks on agriculture. Agricultural households rely on weather for their incomes, and therefore may be vulnerable to changes and unexpected weather events that might affect their production. Thus, we investigate if rainfall shocks affect agricultural and non-agricultural households differently, based on the household main economic activity. More specifically, we estimate a fully interacted models where every control is interacted with an indicator taking the value of 1 if either the biological father or the household head works in agricultural household indicator. The results for both the non-interacted shock variables and the interacted variables are reported in Table 7.

Our results confirm the significant negative effects of the number of rainfall shocks during the first 1,000 days on WM and, in fact, the magnitude of the effects are larger than in the main specification. In addition, we also find significant negative effects on LTM. However, we find no statistically significant differences when comparing the impact on agricultural and non-agricultural households.

INSERT TABLE 7 HERE

7.2. Effects of rainfall shocks on other earlier human-capital accumulation indicators

Since FCSs were only measured during Round 4, we need to rely on other variables to explore how rainfall shocks during early childhood could have affected the children's human-capital accumulation in earlier stages of life (i.e., before Round 4). In this sub-section, we evaluate the effects of rainfall shocks on heights-for-age at ages 5 and 8, to test if there is a nutritional channel through which the

shocks might have impacted on FCS. We also evaluate the effects of shocks on receptive vocabulary test scores, observed in previous rounds of the YLS, which may allow us to find earlier effects on an important domain-specific cognitive skill.

Table 8 presents results on the impact of rainfall shocks on Height-for-Age z-scores (HAZ) at age 5 and 8, and on the PPVT score at the same ages and age 12¹³. Panel A reports the results for shocks during the first 1,000 days, while Panel B presents the results for shocks disaggregated by period and the direction of the shocks.

For HAZ, Panel A shows no effects of total rainfall shocks at any age. However, Panel B shows that there are impacts of abnormally low shocks (drought) on HAZ in childhood, both for pre-birth and postbirth shocks. Also, abnormally high shocks during the post-birth period have significant positive effects. These results suggest there might be a nutritional channel through which rainfall shocks affect FCS. It is possible that we only find these effects at age 5 because early childhood impacts on nutritional status may disappear by mid-childhood, even when FCS deficits prevail for longer.

For PPVT, Panel A shows that there are negative effects both at age 5 and age 8. The effects disappear at age 12. These results coincide with those found by Chang et al. (2022) in the case of rainfall shocks in India. This suggest that the children's cognitive skills are only affected at an early age. Similar to what we find in the case of HAZ, these results might indicate that children are capable of catching-up from the early life negative effects of rainfall shocks on domain-specific skills, such as understanding vocabulary. However, this is not the case for FCS, as we continue to detect effects at age 12, as discussed in Section 6.

INSERT TABLE 8 HERE

7.3. Effects on household's food security

In this sub-section, we explore if the rainfall shocks had an impact on the households' food security. More specifically, we measure the effect of shocks on the self-reported Household Food Insecurity Access Scale (HFIAS). We evaluate the impact of rainfall shocks that occur a year prior to the month of the interview, for each round. A limitation of this strategy is that—given the way the fieldwork was performed—most households within the same community were interviewed during the same month. This reduces the variability of the shocks, as most households within the same community experience the same number of shocks. Due to this limitation, we introduce variation between rounds, within the

¹³ The information for the younger siblings comes from the rounds in which they were of similar ages.

same household. As the HFIAS questions were included in Rounds 3 and 4, we combine both rounds of answers, and control for household fixed effects. As our estimations are at the household level (the level at which these new, self-reported variables are reported), and we are including household fixed effects, we do not include child- or household-level controls.

The results are reported in Table 9. Panel A reports for the effects of the number of shocks a year prior to the interview, and Panel B divides abnormally high shocks from abnormally low shocks, for the same period. Results in Panel A show that shocks in the year prior to the interview have a positive effect on the HFIAS, implying that they increase food insecurity. This is true both for abnormally high and abnormally low shocks, as shown in Panel B. These results are consistent with the hypothesis that rainfall shocks affect households' available income to invest in children's cognitive skills through nutrition.

INSERT TABLE 9 HERE

7.4. The role of JUNTOS

In this section we test if public policies could have a role in lessening or remediating the negative effects of early rainfall shocks on FCS (see Freund et al., 2022). Indeed, Scott et al., (2022) find that receipt of Peru's JUNTOS program (the same program considered here) at a relatively younger age is associated with improvements in IC.

We estimate the impact of the shocks comparing children living in JUNTOS beneficiary households against non-beneficiaries, by estimating a fully interacted model where every independent variable is interacted by two different JUNTOS indicators: One being a dichotomic variable taking the value of one when the child was part of JUNTOS during the first 1,000 days of life, and the other being a dichotomic variable taking the value of one when the child was part of JUNTOS during the child was part of JUNTOS after the first 1,000 days of life. In the first case, the estimated parameter for interaction term would informs us if experiencing a shock while being enrolled in the JUNTOS program could have helped to reduce the effect of the shock (prevention effect).¹⁴ In the second case, the estimated parameter gives information on the role of the conditional cash transfer (CCT) program to remediate the negative effects of rainfall shocks after the first 1,000 days (remediating effect).

¹⁴ Scott et al., (2022) show that children receiving JUNTOS for the first time before the age of 6.7 years experienced a positive impact on Inhibitory Control, relative to those benefiting from the programme at a later age.

Table 10 shows the results for heterogeneous effects for children that are part of the JUNTOS CCT program. We only find significant effects of shocks on IC with no significant evidence of prevention due to JUNTOS, though the coefficient estimate is positive. Our estimate for a possible remediation effect of JUNTOS after the first 1,000 days for IC is significantly positive and completely offsets the negative effects of the shock. Finding that a CCT program like JUNTOS may have a remediation effect for the rainfall shocks serves as further possible evidence of the income mechanism. If rainfall is affecting FCS through household income, this may mean that cash transfers will help households increase their investments in their children and reduce the effects of the shock.

INSERT TABLE 10 HERE

These results, thus, provide some initial evidence of the potential for public policies to remediate the negative impact of rainfall shocks, and are consistent with literature suggesting that FCS might still be malleable at early stages of life. However, given that the JUNTOS program was not randomly allocated, but instead was focused on the "poorest among the poor", we should be careful with how we interpret these results. It could be the case that the results we observe come from the fact that these households are the poorest, and not just for being part of JUNTOS. However, in this case we should not find differences between those children that were part of JUNTOS during the first 1,000 days and those that were part of JUNTOS after the first 1,000 days, as all these children are part of this disadvantaged group.

8. Discussion and further analysis

This paper analyses the long-term impacts of experiencing rainfall shocks during early childhood on a set of foundational cognitive skills (FCS) including inhibitory control (IC), working memory (WM), long-term memory (LM), and implicit learning (IL). We hypothesise that these shocks can have long-lasting negative effects on cognitive development for children in vulnerable contexts, in part through affecting income via different channels. As rainfall has a major impact on agricultural production, we hypothesize that rainfall shocks that influence this activity could affect access to the resources and prices paid for inputs necessary to invest in children's FCS, not only for households that depend directly on agriculture to generate income, but also for non-agricultural households, as reduced agricultural production could affect prices and economic activity in local markets.

By matching novel data on four FCSs from the Young Lives Study with gridded rainfall data at a community level, we are able to use a fixed-effects strategy to identify the possible impacts of these shocks on these cognitive skills. Our analysis shows that rainfall shocks during the first 1,000 days of

life have long-lasting negative effects on WM and IC. We also find negative effects on LTM, under certain conditions, but no evidence of effects on IL. When disaggregating the effects by type of shock and periods of the shock, for WM and IC we find that the in-utero effects are particularly important, only for this period are the point coefficients statistically significant. While point estimates are larger (in absolute values) for droughts compared to floods, statistically, the effects of both type of shocks during this period are undistinguishable.

In terms of agriculture as a possible mechanism, we do not observe any effects that are specific to agricultural households. This could be because it is not only agricultural households that are affected by these shocks, as the shocks can affect other sources of income related to agricultural output such as food prices, food availability, income sources, or access to services. However, we do observe that shocks increase self-reported food insecurity and reduce the nutritional status in childhood, which could indicate that rainfall shocks affect FCS through reductions in investment in human capital when the children are younger and, more specifically, investment in nutrition. We find evidence of early impacts of rainfall shocks on vocabulary skills, measured by the PPVT. These negative effects disappear by late childhood or early adolescent, which contrasts with the long-lasting effects on FCS found in our main analysis.

Additionally, by analysing heterogeneous effects for JUNTOS recipients, our results suggest that there may be space for public policies to remediate the negative effects of rainfall shocks on FCSs. However, we found no evidence of JUNTOS working as a preventative measure. Thus, the role of the program seems to be limited to remediating effects, possibly indicating that households are failing to optimally allocate resources over time. These results strengthen the hypothesis that the rainfall shocks affect the FCSs through decreasing households' income: That a cash transfer can decrease the negative effect of the shocks indicates that a main mechanism must be related to income.

In conclusion, our results shed light on the negative effects of early exposure to rainfall shocks on the long-term formation of children's WM and IC. Our main hypothesis states that rainfall shocks reduce agricultural output and, therefore, reduce resources and increase prices of the inputs required for investing in children's cognitive skills. Under this model, the main mechanism driving our results should be related to the rainfall shocks decreasing households' abilities to invest in children's human capital. This study provides some evidence suggesting that this might be the case.

However, our data and methodology do not allow us to directly link our results on the mechanisms to our results on FCSs. Moreover, there could be other mechanisms through which rainfall shocks affect FCSs that we have not tested in this study. Future studies should focus on understanding better the mechanisms behind the relationship between early exposure to rainfall shocks and FCS formation, as they could help to develop better prevention policies to mitigate the negative effects of these shocks in the context of climate change and poverty.

With climate change well underway, unpredictable rainfall shocks are becoming more common. These changes tend to disproportionately affect the most vulnerable households, especially those from LMICs. In contexts where inequalities are high, and many struggle to escape poverty, giving these communities the right tools to overcome more frequent weather shocks will help prevent their children from losing future human capital. Moreover, the international community must strengthen efforts to reduce global warming and, with it, the severity and persistence of these shocks.

References

- Alemu, T., Asgedom, G., Liebenberg, J., Mekonnen, A., Seager, J., Tefera, B., Woldehanna, T. (2003). Young Lives Preliminary Country Report: Ethiopia.
- Alderman H, Hoddinott J, Kinsey B (2006). Long term consequences of early childhood malnutrition. Oxf Econ Pap 58:450–474.
- Almond, D., Currie, J., Duque, V. (2018). Childhood circumstances and adult outcomes: Act II. Journal of Economic Literature, 56(4), 1360–1446.
- Amare, M., Jensen, N. D., Shiferaw, B., Cissé, J. D. (2018). Rainfall shocks and agricultural productivity: Implication for rural household consumption. Agricultural systems, 166, 79-89.
- Ardila, A., Rosselli, M., Matute E. (2005). The influence of the parents educational level on the development of executive functions. Developmental Neuropsychology, 28(1): p. 539-560.
- Baez, J.E., De la Fuente A., Santos I.V. (2009). Do natural disasters affect human capital? An assessment based on existing empirical evidence. IZA Discussion Paper, Bonn, Germany: Institute for the Study of Labour. 5164.
- Behrman, J.R., Briones, K., Cueto, S., Favara, M., Freund, R, Hittmeyer, A., Lopez, J., Pazos, N., Sánchez, A., Scott, D., Sheridan, M., Woldehanna, T. (2022). Measuring foundational cognitive skills in Young Lives using RACER. (Young Lives Technical Note 54). Young Lives.
- Berhane, G., D. O. Gilligan, J. Hoddinott, N. Kumar, A.S. Taffesse (2014). Can social protection work in Africa? the impact of Ethiopias Productive Safety Net Programme. Economic Development and Cultural Change, 63(1): p. 1-26.
- Black, M. M., Walker, S.P., Fernald, L., Andersen, C.T., DiGirolamo, A.M., Lu C., McCoy D.C., Fink, G., Shawar Y.R., Shiffman, J. (2017). "Early Childhood Development Coming of Age: Science through the Life Course." *The Lancet* 389, no. 10064: 77-90.
- Blair, C. Razza, R.P. (2007), Relating effortful control, executive function, and false belief understanding to emerging math and literacy ability in kindergarten. Child Development, 78(2), p. 647-663.
- Blair, C. (2002), School readiness: integrating cognition and emotion in a neurobiological conceptualization of children's functioning at school entry. Am. Psychol., 57(2), p. 111-127.
- Chang, G., Favara, M., Novella, R. (2022). The origins of cognitive skills and non-cognitive skills: The long-term effect of in-utero rainfall shocks in India. Economics & Human Biology, 44, 101089.
- Cornwell, K., Inder, B. (2015). Child health and rainfall in early life. The Journal of Development Studies, 51(7), 865-880.
- Dercon, S. (2004). Growth and Shocks: Evidence from Rural Ethiopia. Journal of Development Economics 74(2): 309–29.
- Dercon, S., & Porter, C. (2014). Live aid revisited: long-term impacts of the 1984 Ethiopian famine on children. Journal of the European Economic Association, 12(4), 927-948.
- Dimitrova, A., Muttarak, R. (2020). After the floods: Differential impacts of rainfall anomalies on child stunting in India. Global Environmental Change, 64.
- Escobal, J., Lanata, C., Madrid, S., Penny, M., Saavedra, J., Suárez, P., Verastegui, H., Villar, E., Huttly, S. (2003). Young Lives preliminary country report: Peru
- Escobal, J., Flores, E. (2008). An assessment of the Young Lives sampling approach in Peru.
- Farah, M., D. Sherab, J. Savagea, L. Betancourta, J. Giannettac, N. Brodskyc, E. Malmudc, H. Hurtc (2006). Childhood poverty: specific associations with neurocognitive development. Brain Research, 1110(1), p. 166-174.
- Felkner, J., Tazhibayeva, K., Townsend, R. (2009). Impact of climate change on rice production in Thailand. American Economic Review, 99(2), 205-10
- Fernald, L., P. Kariger, P. Engle, A. Raikes (2009), Examining early child development in lowincome countries: a toolkit for the assessment of children in the first five years of life. Technical Report, The World Bank.
- Ford, C., Yeon Kim, H., Brown, L., Abers, L., Sheridan, M. (2019). A cognitive assessment tool designed for data collection in the field in low- and middle-income countries. Research in Comparative and International Education, I-16.

- Freund, R., Favara, M., Porter, C, Behrman, J.R. (2022). Public Works Programmes and Foundational Cognitive Skills: The Case of the Ethiopian Productive Safety Net Programme. University of Pennsylvania Population Center Working Paper (PSC/PARC), 2022-99.
- Glewwe, P. and E.A. Miguel (2007). The Impact of Child Health and Nutrition on Education in Less Developed Countries, in T. Schultz and J. Strauss (eds.) Handbook of Development Economics, Volume 4, 3561–3606, Amsterdam: North Holland.
- Grabrucker, K., Grimm, M. (2021). Is There a Rainbow after the Rain? How Do Agricultural Shocks Affect Non-Farm Enterprises? Evidence from Thailand. American Journal of Agricultural Economics, 103(5), 1612-1636.
- Hamoudi, A., Sheridan, M. (2015). Unpacking the black box of cognitive ability: A novel tool for assessment in a population-based survey [unpublished manuscript]. Available from: http://thew.web.unc.edu/files/2013/08/hamoudi.pdf.
- Hirvonen, K., Sohnesen, T. P., Bundervoet, T. (2020). Impact of Ethiopia's 2015 drought on child undernutrition. World Development, 131, 104964.
- Hyland, M., Russ, J. (2019). Water as destiny–The long-term impacts of drought in sub-Saharan Africa. World Development, 115, 30-45.
- Jayachandran, S. (2006). Selling labor low: Wage responses to productivity shocks in developing countries. Journal of political Economy, 114(3), 538-575.
- Klenberg, L., M. Korkman, P. Lahti-Nuuttila (2001), Differential development of attention and executive functions in 3 to 12 year old Finnish children. Developmental Neuropsychology, 20(1): p. 407-428.
- Lezak, M. D. (1982). The problem of assessing executive functions. International journal of Psychology, 17(1-4), 281-297.
- Lloyd-Hughes, B., Saunders, M. A. (2002). A drought climatology for Europe. *International Journal* of Climatology: A Journal of the Royal Meteorological Society, 22(13), 1571-1592.
- Lu, Chunling, Maureen M. Black, and Linda M. Richter. (2016). Risk of Poor Development in Young Children in Low-Income and Middle-Income Countries: An Estimation and Analysis at the Global, Regional, and Country Level. *The Lancet Global Health*. Accessed 2016/11/06. <u>http://dx.doi.org/10.1016/s2214-109x(16)30266-2</u>.
- Lupien, S.J., B. S. McEwen, M. R. Gunnar, C. Heim. (2009). Effects of stress throughout the lifespan on the brain, behavior and cognition. Nature Reviews: Neuroscience, 10(6): p. 434-445.
- Maccini, S, Yang, D. (2009). Under the Weather: Health, Schooling, and Economic Consequences of Early-Life Rainfall. *American Economic Review* 99, no. 3: 1006-26. <u>http://dx.doi.org/doi</u>: 10.1257/aer.99.3.1006.
- Maluccio, J.A., Hoddinott, J., Behrman J.R., Martorell, R., Quisumbing, A.R., Stein, A.D. (2009). The Impact of Improving Nutrition During Early Childhood on Education among Guatemalan Adults. *The Economic Journal* 119, no. 537 (2009): 734-63. http://www.jstor.org/stable/20485342.
- Matsuura, K., Willmott, C. J. (2015). Terrestrial precipitation: 1900–2014 gridded monthly time series. University of Delaware Dept. of Geography.[Available online at http://climate.geog.udel.edu/~ climate/html_pages/Global2014/README. GlobalTsP2014. html].
- McLaughlin, K.A., Sheridan, M.A., Lambert, H.K. (2014), Childhood adversity and neural development: Deprivation and threat as distinct dimensions of early experience. Neuroscience Biobehavioral Reviews, 47(November): p. 578-591.
- Nelson, C.A., Sheridan, M.A. (2011), Lessons from neuroscience research for understanding causal links between family and neighborhood characteristics and educational outcomes. In G.J. Duncan & R.J. Murnane (Eds.), Whither opportunity? Rising inequality, schools, and children's life chances. New York, NY: Russell Sage Foundation, p. 27-46.
- Nicholas, K., Campbell, L., Paul, E., Skeltis, G., Wang, W., Gray, C. (2021). Climate anomalies and childhood growth in Peru. Population and Environment, 43(1), 39-60.
- Noble, K., M. Norman, Farah, M. (2005). Neurocognitive correlates of socioeconomic status in kindergarten children. Developmental Science, 8(1), p. 74-87.
- Noble, K., McCandliss B., Farah, M. (2007). Socioeconomic gradients predict individual differences in neurocognitive abilities. Developmental Science, 10(4), p. 464-480.

- Randell, H., Gray, C. (2019). Climate change and educational attainment in the global tropics. PNAS, 116(18).
- Randell, H., Gray, C., Grace, K. (2020). Stunted from the start: Early life weather conditions and child undernutrition in Ethiopia. Social Science & Medicine, 261, 113234.
- Rijkers, B., Söderbom, M. (2013). The effects of risk and shocks on non-farm enterprise development in rural Ethiopia. World Development, 45, 119-136.
- Rocha, R., Soares, R. R. (2015). Water scarcity and birth outcomes in the Brazilian semiarid. Journal of Development Economics, 112, 72-91.
- Rosales-Rueda, M. (2018). The impact of early life shocks on human capital formation: Evidence from El Niño floods in Ecuador. Journal of health economics, 62, 13-44.
- Sánchez, A., Escobal, J. (2020). Survey attrition after 15 years of tracking children in four developing countries: the Young Lives study. Review of Development Economics, 24(4), 1196-1216.
- Sánchez, A., Favara, M., Sheridan, M., Behrman, J. 2022. Impact of early nutritional status on foundational cognitive skills: evidence from two developing countries. Mimeo.
- Scott, D., Lopez, J., Sánchez, A., Behrman, J.R. (2022). The Impact of the JUNTOS Conditional Cash Transfer Programme on Foundational Cognitive Skills: Does Age of Enrollment Matter? University of Pennsylvania Population Center Working Paper (PSC/PARC), 2022-98.
- Shah, M., Steinberg, B. M. (2017). Drought of opportunities: Contemporaneous and long-term impacts of rainfall shocks on human capital. Journal of Political Economy, 125(2), 527-561.
- Sheridan M. A., K.A. McLaughlin (2014), Dimensions of early experience and neural development: deprivation and threat. Trends in Cognitive Sciences, 18(11): p. 580-585.
- Sheridan M. A., Sarsour K., Jutte D., D'Esposito M., Boyce W.T. (2012). The impact of social disparity on prefrontal function in childhood. PLoS ONE 7(4).
- Sheridan, M.A., J. How, M. Araujo, M. A. Schamberg, C.A. Nelson (2013). What are the links between maternal social status, hippocampal function, and HPA axis function in children? Developmental Science, 16(5), p. 665–675.
- Shonko, J.P., A.S. Garner (2012), The lifelong effects of early childhood adversity and toxic stress. Pediatrics, 129(1), p. 232-246.
- Skoufias, E., Vinha, K. (2012). Climate variability and child height in rural Mexico. Economics & Human Biology, 10(1), 54-73.
- Thai, Thuan Q., Evangelos M. Falaris. (2014). Child Schooling, Child Health, and Rainfall Shocks: Evidence from Rural Vietnam. The Journal of Development Studies 50, no. 7: 1025-37. <u>http://dx.doi.org/10.1080/00220388.2014.903247</u>.
- Todd, P., K, Wolpin, (2003). On the specification and estimation of the production function for cognitive achievement. Economic Journal, 113, p. F3-F33.
- USGCRP (U.S. Global Change Research Program). (2017). Climate science special report: Fourth National Climate Assessment, volume I. Wuebbles, D.J., D.W. Fahey, K.A. Hibbard, D.J. Dokken, B.C. Stewart, and T.K. Maycock, eds. <u>https://science2017.globalchange.gov</u>. doi:10.7930/J0J964J6.
- Victora, C.G., Adair, L., Fall, C., Hallal, P.C., Martorell, R., Richter, L., Singh Sachdev H. (2008). Maternal and Child Undernutrition: Consequences for Adult Health and Human Capital. *The Lancet* 371, no. 9609: 340-57. <u>http://dx.doi.org/10.1016/s0140-6736(07)61692-4</u>.
- Victora, C.G., de Onis, M., Curi, P., Blössner, M., Shrimpton, R. (2010) Worldwide Timing of Growth Faltering: Revisiting Implications for Interventions. *Pediatrics* 125: e473-e80. <u>http://pediatrics.aappublications.org/content/early/2010/02/15/peds.2009-1519.abstract</u>.

	Mean	Std. Dev.
Households		
Households in urban areas in Round 1 (%)	67.57	46.82
Household main activity is agriculture (%)	52.61	49.94
Mother's schooling attainment (grades)	7.56	4.40
Mother's primary language is Spanish	70.40	45.66
Regions:		
Coast	46.92	49.92
Mountain	34.35	47.50
Jungle	18.73	39.03
Sample size	1,511	
Index children		
Female (%)	49.83	50.01
Age (years) in round 3 (2009)	7.9157	.2996
Age (years) in round 4 (2013)	11.915	.30342
Height-for-age (Z-score) in 2009	-1.15	1.04
Oldest sibling (first born)		
Sample size	1,511	
Younger Siblings		
Female (%)	50.41	50.04
Age (years) in round 3 (2009)	4.73	1.27
Age (years) in round 4 (2013)	8.74	1.28
Height-for-age (Z-score) in 2006	-1.52	1.51
Sample size	605	

Table 1. Average sample characteristics

Table 2. Average foundational cognitive skills: baseline	tasks and cha	allenge tasks
	Baseline	Challenge
	tasks	tasks
Inhibitory control (IC)- Inverse of standardized		
combined average response time & accuracy		
Average	.187	.097
Index children	.210	.121
Younger siblings	.131	.038
Working momory (WM) Inverse of standardized		
average response time		
Average	- 126	113
Index children	- 126	117
Younger siblings	126	.102
ŭ		
Long-term memory (LTM)- Standardized proportion		
of correct answers at first touch		
Average	.004	.056
Index children	.024	.082
Younger siblings	044	007
Implicit learning (II.)- Inverse of standardized		
average response time		
Average	008	.139
Index children	030	.127
Younger siblings	.048	.170

Period of	Type of shock	Average number of shocks				
shock	Type of shock	All children	Index children	Siblings		
	All	0.48	0.57	0.29		
F *4		(0.875)	(0.91)	(0.761)		
First	Abnormally high	0.36	0.39	0.28		
luuu days		(0.758)	(0.76)	(0.756)		
uays	Abnormally low	0.13	0.18	0.01		
		(0.480)	(0.56)	(0.107)		
	All	0.18	0.24	0.06		
		(0.399)	(0.44)	(0.252)		
In utero	Abnormally high	0.12	0.15	0.06		
period		(0.331)	(0.36)	(0.248)		
	Abnormally low	0.06	0.09	0.00		
		(0.237)	(0.28)	(0.044)		
	All shocks	0.30	0.33	0.23		
Post-		(0.534)	(0.52)	(0.563)		
birth	Abnormally high	0.23	0.24	0.22		
(Years 1		(0.485)	(0.45)	(0.560)		
and 2)	Abnormally low	0.07	0.09	0.01		
		(0.248)	(0.29)	(0.076)		

Table 3: Prevalence of SPI shocks (2.0. SD) during rainy seasons

Period	Type of shock	% of chil le	dren experi ast one shoc	encing at ek	Average number of shocks, fo those experiencing at least on shock		
OI SHOCK		All children	Index children	Siblings	All children	Index children	Siblings
First	All	27.0 %	31.0 %	16.9 %	1.8	1.8	1.7
1000	Abnormally high	20.9 %	22.9 %	16.3 %	1.7	1.7	1.3
days	Abnormally low	6.6 %	9.2 %	0.6 %	1.9	1.9	1.7
In utero	All	17.9 %	23.0 %	6.1 %	1.0	1.0	1.0
period	Abnormally high	12.3 %	15.1 %	6.0 %	1.0	1.0	1.0
	Abnormally low	6.0 %	8.5 %	0.2 %	1.0	1.0	1.0
Post-	All shocks	27.0 %	31.4 %	16.9 %	1.1	1.1	1.3
birth	Abnormally high	20.9 %	22.9 %	16.3 %	1.1	1.0	1.3
(Years 1 and 2)	Abnormally low	6.6 %	9.2 %	0.6 %	1.0	1.0	1.0

Table 4: Prevalence of SPI shocks (2	2.0. SD)	during rainy seas	sons, number of shocks
--------------------------------------	----------	-------------------	------------------------

Table 5: Main Results						
Dependent variable:	Inhibitory Control	Working Memory	Long Term Memory	Implicit Learning		
	(1)	(2)	(3)	(4)		
Number of shocks during the first 1,000 days	-0.030	-0.066***	-0.032	0.025		
	(0.024)	(0.022)	(0.025)	(0.023)		
Adjusted R2	0.371	0.068	0.482	0.321		
Number of observations	1,714	1,708	1,705	1,711		

Note: All coefficients are standardized. Controls included: community fixed effects, year and month-of-birth fixed effects, sex, the native tongue of the mother, highest education level acquired by parents, whether the child was born prematurely, performance in the baseline tasks, whether the house is an agricultural household, whether the task was administered during the weekend, and the time of the day when the tasks were administered. Standard errors (reported in parentheses) are clustered at cluster level. * p < 0.1 * p < 0.05 ** p < 0.01.

Dependent variable:	Inhibitory Control	Working Memory	Long Term Memory	Implicit Learning
	(1)	(2)	(3)	(4)
Number of abnormally high shocks during in-utero period	-0.104*	-0.136	-0.173	-0.061
	(0.059)	(0.148)	(0.107)	(0.064)
Number of abnormally low shocks during in-utero period	-0.308*	-0.300***	-0.222	0.280
	(0.166)	(0.104)	(0.221)	(0.477)
Number of abnormally high shocks during post-birth period	0.031	-0.044	0.016	0.076
	(0.044)	(0.088)	(0.039)	(0.061)
Number of abnormally low shocks during post-birth period	0.152	0.250	0.317	-0.226
	(0.184)	(0.267)	(0.271)	(0.411)
Adjusted R2	0.372	0.067	0.483	0.320
Number of observations	1,714	1,708	1,705	1,711
F-tests (p-values)				
In utero low = In utero high shocks	0.224	0.430	0.840	0.517
Post birth low = Post birth high shocks	0.521	0.372	0.285	0.503
In utero low = Post birth low shocks	0.193	0.146	0.286	0.576
In utero high = Post birth high shocks	0.140	0.689	0.196	0.266

Table 6: Main results, disaggregated between abnormally low and abnormally high rainfall shocks and between in-utero and post-birth periods

Note: All coefficients are standardized. Controls included: community fixed effects, year and month-ofbirth fixed effects, sex, the native tongue of the mother, highest education level acquired by parents, whether the child was born prematurely, performance in the baseline tasks, whether the house is an agricultural household, whether the task was administered during the weekend, and the time of the day when the tasks were administered. Standard errors (reported in parentheses) are clustered at cluster level. * p<0.1 * p<0.05 ** p<0.01.

Dependent variable:	Inhibitory Control	Working Memory	Long Term Memory	Implicit Learning
	(1)	(2)	(3)	(4)
Number of shocks during the first 1,000 days	-0.042	-0.105**	-0.069**	0.015
	(0.035)	(0.046)	(0.034)	(0.032)
Number of shocks during the first 1,000 days * Agricultural household	0.005	0.040	0.028	0.002
	(0.040)	(0.050)	(0.028)	(0.054)
Number of observations	1,714	1,708	1,705	1,711
Adjusted R2	0.377	0.063	0.495	0.325

Table 7: Results for agricultural households, whole sample (4 months of rainy season)

Note: All coefficients are standardized. Controls included: community fixed effects, year and month-ofbirth fixed effects, sex, the native tongue of the mother, highest education level acquired by parents, whether the child was born prematurely, performance in the baseline tasks, whether the house is an agricultural household, whether the task was administered during the weekend, and the time of the day when the tasks were administered. Standard errors (reported in parentheses) are clustered at cluster level. * p<0.1 * p<0.05 ** p<0.01.

Table 8: Effects on Height-for-age and PPVT

Dependent variable:	Age 5 Height for age z- score	Age 8 Height for age z- score	Age 5 PPVT	Age 8 PPVT	Age 12 PPVT
	(1)	(2)	(3)	(4)	(5)
Panel A: First 1000 days					
Number of shocks during the first 1,000 days	-0.080	-0.002	-1.101***	-0.874***	-0.188
	(0.055)	(0.047)	(0.374)	(0.325)	(0.578)
Adjusted R2	0.228	0.276	0.594	1,662	1,688
Panel B: By period					
Number of abnormally high shocks during in-utero period	0.160	0.081	-1.962	2.245	-0.137
	(0.146)	(0.137)	(1.329)	(1.772)	(0.867)
Number of abnormally low shocks during in-utero period	-0.206*	-0.036	-0.154	-2.500**	-0.166
	(0.113)	(0.101)	(0.680)	(0.988)	(1.049)
Number of abnormally high shocks during post birth period	0.392*	0.064	-0.812	-3.175	-6.426***
	(0.230)	(0.168)	(3.830)	(5.795)	(1.448)
Number of abnormally low shocks during post birth period	-0.599***	-0.130	-3.576*	0.535	5.561*
	(0.085)	(0.116)	(1.898)	(5.675)	(2.932)
Adjusted R2	0.229	0.275	0.593	0.402	0.367
Number of observations	1,709	1,712	1,523	1,662	1,688
F-tests (p-values)					
In utero low = In utero high shocks	0.322	0.944	0.789	0.415	0.004
Post birth low = Post birth high shocks	0.014	0.571	0.102	0.598	0.091
In utero low = Post birth low shocks	0.002	0.452	0.629	0.749	0.008
In utero high = Post birth high shocks	0.138	0.590	0.279	0.081	0.987

Note: All coefficients are standardized. Controls included: community fixed effects, year and month-of-birth fixed effects, sex, the native tongue of the mother, highest education level acquired by parents, whether the child was born prematurely, performance in the baseline tasks, whether the house is an agricultural household, whether the task was administered during the weekend, and the time of the day when the tasks were administered. Standard errors (reported in parentheses) are clustered at cluster level. * p<0.1 * p<0.05 ** p<0.01.

	HFIAS (R2 AND R3)
Dependent variable:	(1)
Panel A: All shocks	(1)
Number of shocks (in year prior to interview)	0.360**
Adjusted R2	(0.152) 0.249
Panel B: Type of shock	
Number of abnormally high shocks (in year prior to interview)	0.429***
	(0.119)
Number of abnormally low shocks (in year prior to interview)	0.045***
	(0.014)
Adjusted R2	0.249
Number of observations	2,656

Table 9: Results on Household level food security

Note: All coefficients are standardized. Controls included: community fixed effects, year and monthof-birth fixed effects, sex, the native tongue of the mother, highest education level acquired by parents, whether the child was born prematurely, performance in the baseline tasks, whether the house is an agricultural household, whether the task was administered during the weekend, and the time of the day when the tasks were administered. Standard errors (reported in parentheses) are clustered at cluster level. * p<0.1 * p<0.05 ** p<0.01.

Table 10: The role of JUNTOS

Dependent variable:	Inhibitory Control	Working Memory	Long Term Memory	Implicit Learning
	(1)	(2)	(3)	(4)
Panel B: Peru				
Number of shocks during the first 1,000 days	-0.071**	-0.046	-0.041	-0.006
	(0.036)	(0.045)	(0.036)	(0.021)
Number of shocks during the first 1,000 days * JUNTOS during first 1,000 days	0.078	0.019	0.025	0.135
	(0.087)	(0.101)	(0.145)	(0.149)
Number of shocks during the first 1,000 days * JUNTOS after first 1,000 days	0.081*	-0.070	0.003	0.064
	(0.047)	(0.076)	(0.040)	(0.051)
Adjusted R2	0.379	0.071	0.513	0.326
Number of observations	1,714	1,708	1,705	1,711

Note: All coefficients are standardized. Controls included: community fixed effects, year and month-of-birth fixed effects, sex, the native tongue of the mother, highest education level acquired by parents, whether the child was born prematurely, performance in the baseline tasks, whether the house is an agricultural household, whether the task was administered during the weekend, and the time of the day when the tasks were administered. Standard errors (reported in parentheses) are clustered at cluster level. * p<0.1 * p<0.05 ** p<0.01.